Dear Prof. Wang,

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. We have addressed the comments raised by the reviewers. Point by point responses to the reviewers’ comments are listed below this letter. In addition, the format of the manuscript has been updated according to the Guidelines and Requirements for Manuscript Revision and the Format for Manuscript Revision. We hope the manuscript would be published soon.

Yours sincerely,
Xu-Guang Chen

Replies to Reviewer #1
Please correct all your minor grammar errors and typos.
Answer: We have rechecked and asked AJE to edit the language.

Please mention the Guidelines in the main text that you use in your routine clinical practice
Answer: We have written in the text guidelines for reference in clinical practice.

It might be nice to see at least more clinical data
Answer: We have added clinical data such as ECG, CT, cardiac ultrasound, and lab test results.

If you mention any other cases, why there are no references?
Answer: We list references where other cases are mentioned.

What have we suppose to learn after reading of your article?
Answer: We have rewritten the discussion section and again summarized what we want to tell our readers.

Replies to Reviewer #2:
This case report is very interesting with clinical impact on management of acute vascular emergency.
Answer: Thank you very much!

Replies to Science editor:
1. The author should supplement the figure legend in more detail.
Answer: We added ECG and CT images.

2. More clinical data should be added when the case is described.
Answer: We have added clinical data such as ECG, CT, cardiac ultrasound, and lab test results.
Replies to Company editor-in-chief:

have sent the manuscript to the author(s) for its revision according to the Peer-Review Report. The author(s) must provide the Signed Consent for Treatment Form(s) or Document(s). Before final acceptance, uniform presentation should be used for figures showing the same or similar contents.

Answer: The format of the manuscript has been updated according to the Guidelines and Requirements for Manuscript Revision and the Format for Manuscript Revision. We have re-arranged the pictures.